Author's response to reviews

**Title:** Potential relevance of quality of life questionnaires to identify candidates for surgical treatment of genital prolapse: a pilot study

**Authors:**

Christian Chauvin (chx2fr@yahoo.fr)
Elisabeth Chéreau (elisabeth.chereau@tnn.aphp.fr)
Marcos Ballester (marcos.ballester@tnn.aphp.fr)
Emile Daraï (emile.darai@tnn.aphp.fr)

**Version:** 2  **Date:** 8 September 2011

**Author's response to reviews:** see over
To
The Editor
BMC Urology

Dear sir,

Enclosed please find our paper:

“Potential relevance of quality of life questionnaires to identify candidates for surgical treatment of genital prolapse: a pilot study.”

By Christian Chauvin, Elisabeth Chéreau, Marcos Ballester, Emile Darai.

The aim of this pilot study was to evaluate prolapse-related symptoms, quality of life and sexuality of patients with validated questionnaires before and after surgery for genital prolapse and assess relevance of such an evaluation to select women for surgery.

We believe that this study could therefore be of interest for the readership of your journal.

None of the authors has potential conflict of interest.
All authors fulfilled all conditions required for authorship and approved the submission.

Each author contribution to the manuscript is listed below:

- Protocol/project development: ED/ EC
- Data collection or management: CC/ MB
- Data analysis: EC/MB/ED
- Manuscript writing/editing: CC/EC/ED

We are looking forward to hearing from you at your earliest convenience concerning your decision about this manuscript.

Sincerely Yours,

Dr Elisabeth Chéreau